**Author’s response to reviews**

**Title:** Measurement of maternal functioning during pregnancy and postpartum: findings from the cross-sectional WHO Pilot Study in Jamaica, Kenya, and Malawi

**Authors:**

Jenny A CRESSWELL (cresswellj@who.int)

Kelli D. Barbour (kellidbarbour@gmail.com)

Doris Chou (choud@who.int)

Affette McCaw-Binns (affette.mccawbinns@uwimona.edu.jm)

Veronique Filippi (veronique.filippi@lshtm.ac.uk)

Jose Guilherme Cecatti (cecatti@unicamp.br)

Maria Barreix (barreixm@who.int)

Max Petzold (max.petzold@gu.se)

Nenad Kostanjsek (kostanjsekn@who.int)

Sara Cottler-Casanova (sara.cottler@me.com)

Lale Say (sayl@who.int)

**Version:** 1 **Date:** 22 Oct 2019

**Author’s response to reviews:**

Dear Editor,

We have corrected the manuscript as detailed in the submission e-mail.

Kind regards,

Jenny CRESSWELL